Author’s response to reviews

Title: Is it the decision of women to choose a cesarean section as the mode of delivery? A review of literature on the views of stakeholders

Authors:

Alice Yuen Loke (hsaloke@polyu.edu.hk)

Louise Davies (louise.davis@polyu.edu.hk)

Yim-wah Mak (yw.mak@polyu.edu.hk)

Version: 6 Date: 26 Jul 2019

Author's response to reviews:

1. Main text of the manuscript is in-line with the submission guidelines for a debate article. it include the subheadings: Background, Main Text and Conclusions.

2. The revised Abstract is submitted in the submissions system.

3. "Key Messages" is removed and now include under the Conclusion section: in the 2nd, 3rd, and 4th paragraphs of the manuscript.

4. The first line of the abstract "this is a literature review" is removed as suggested.

5. the manuscript is a clean copy and has been proof read, there is no tracked changes , colored text, or highlighting.